Author’s response to reviews

Title: Is it the decision of women to choose a cesarean section as the mode of delivery? A review of literature on the views of stakeholders

Authors:

Alice Yuen Loke (hsaloke@polyu.edu.hk)

Louise Davies (louise.davis@polyu.edu.hk)

Yim-wah Mak (yw.mak@polyu.edu.hk)

Version: 5 Date: 24 Jul 2019

Author's response to reviews:

AYL, and YWM made substantial contributions to the conception and design of this review. LD searched for literature from electronic data bases and developed the summary table. AYL and LD worked together to analyze and interpret the findings. LD was responsible for drafting the manuscript. AYL and YWM provided the intellectual and critical approval of the final manuscript for publication. AYL and YWM were the supervisors of LD at the time of this review was conducted.